Author's response to reviews

Title: Distribution of immunodeficiency fact files with XML - from Web to WAP

Authors:

Jouni Valiaho (jouni.valiaho@uta.fi)
Pentti Riikonen (pentti.riikonen@it.utu.fi)
Mauno Vihinen (mauno.vihinen@uta.fi)

Version: 5 Date: 20 May 2005

Author's response to reviews: see over
Dear Editor,

we hereby submit a revised version of our manuscript entitled “Distribution of immunodeficiency fact files – from Web to WAP” (MS: 5876490813157291). We have taken into account the formatting changes requested: Competing interests - We have added text “The authors declare that they have no competing interests”.
References – We have formatted the references in the journal's style. The reference list contains all authors' names.
Tables – We have changed the borders of tables to visible black lines.
Figure 1 – We have reformatted the figure 1 by making it shorter.
Additional files - We have added a reference to the additional files in the text.

Yours sincerely,

Jouni Väliaho,
Mauno Vihinen